Author's response to reviews

Title: Differential expression of a new isoform of DLG2 in renal oncocytoma

Authors:

Dmitry Zubakov (genosys@gmx.de)
Zorica Stupar (stupar_zorica@yahoo.de)
Gyula Kovacs (gyula.kovacs@urz.uni-heidelberg.de)

Version: 3 Date: 25 April 2006

Author's response to reviews: see over
Dear Editor,
we are delighted upon your decision to accept our manuscript "Differential expression of a new isoform of DLG2 in renal oncocytoma" for publication in BMC Cancer.
Please, find below formatting changes we did in a final version of the manuscript according to the requests of BioMed Central Editorial Production Team:

1) website address is moved from the text of the manuscript to the reference list;
2) references are changed according to the recommendations. EndNote program with BMC Cancer style template was used;
3) one duplicate reference (former #9) was removed from the list, citations in the text were changed respectively;
4) table is included in the manuscript after figure legends;
5) all figures are cited in the text.

With best regards,
Prof. Gyula Kovacs